Author's response to reviews

Title: The case study approach

Authors:

Sarah Crowe (sarah.crowe@nottingham.ac.uk)
Kathrin Cresswell (kathrin.beyer@ed.ac.uk)
Ann Robertson (A.R.R.Robertson@ed.ac.uk)
Guro Huby (guro.huby@ed.ac.uk)
Anthony Avery (anthony.avery@nottingham.ac.uk)
Aziz Sheikh (aziz.sheikh@ed.ac.uk)

Version: 6 Date: 31 May 2011

Author's response to reviews: see over
26th May 2011

Dear Adrian,

Re: 4726512534893146/The case study approach

Thank you for your email of 13th May 2011 provisionally accepting our paper. We have made a few further relatively minor revisions in the light of the feedback from Reviewer 2. Specifically, we have included a sub-section on ‘main limitations’ in relation to each of our case studies (see Boxes 1-4). We have, furthermore, made clear that we do not have any conflicts of interest in relation to this paper.

We trust that these revisions are to your satisfaction and we look forward to seeing this published in due course. Please do not hesitate to contact us if you require any further information or clarification.

With kind regards,

Sarah Crowe
Research Fellow
University of Nottingham

&

Aziz Sheikh
Professor of Primary Care Research & Development
The University of Edinburgh
On behalf of the co-authors